Learning and Representation of Planning Ability in the Human Brain: Executive Function and Decision Making

by

Keita Mitani

Student ID Number: 1196017

A dissertation submitted to the Engineering Course, Department of Engineering, Graduate School of Engineering, Kochi University of Technology, Kochi, Japan

in partial fulfillment of the requirements for the degree of Doctor of Engineering

Assessment Committee: Supervisor: Yukinobu Hoshino Co-Supervisor: Xiangshi Ren Co-Supervisor: Hiroshi Kadota Hiroaki Shigemasu Kyoko Shibata

March 2023

Abstract

In this dissertation, I describe the three investigated results that brain representations are involved with decision-making. Firstly, we examined a transition of brain activation in decision-making at problem-solving by long-term learning. The methodology of this study involves the Tower of Hanoi (ToH) to investigate executive functions related to the learning process. Generally, ToH is used to measure baseline performance, learning rate, offline learning, and transfer. However, this study performs experiments on long-time learning effects for ToH solving. The participants were involved in learning the task over seven weeks. Learning progress was evaluated based on improvement in performance and correlations with the learning curve. All participants showed a significant improvement in planning and decision-making over seven weeks of time duration. Brain activation results from functional magnetic resonance imaging (fMRI) showed a statistically significant decrease in the activation degree in the dorsolateral prefrontal cortex, parietal lobe, inferior frontal gyrus, and premotor cortex between before and after learning. Our pilot study showed that updating information and shifting issue rules were found in the frontal lobe. Through monitoring performance, we can describe the effect of long-time learning initiated at the frontal lobe and then convert it to a task execution function by analyzing the frontal lobe maps. This process can be observed by comparing the learning curve and the fMRI maps. It was also clear that the degree of activation tends to decrease with the number of tasks, such as through the midphase and the end-phase of training. The elucidation of this structure is closely related to decision-making in human behavior, where brain dynamics differ between "thinking and behavior" during complex thinking in the early stages of training and instantaneous "thinking and behavior" after sufficient training. Since this is related to human learning, elucidating these mechanisms will allow the construction of a brain function map model that can be used universally for all training tasks. Next, we focused on brain activation during the decision-making process, which was selected based on a single evaluation criterion. Neuro-marketing, which explores the human decision-making process, is attracting attention by measuring the reaction of the brain in an objective way, and taking advantage of progress in brain science and psychophysics. Numerous studies have been conducted to date on preference-based decision-making, and the evaluation criteria of the decision-making process are diverse, such as food preferences, product costs, or facial features. In this research, decision-making experiments by several evaluation criteria were conducted using fMRI scanner in order to elucidate the brain regions comprehensively involved in decision-making and specific brain regions related to each evaluation criterion. This experiment measured brain activity during paired comparison tasks based on a single evaluation criterion by fMRI, and attempted to identify specific brain regions related to decision-making based on each evaluation criterion. Participants performed a decision-making task that involved choosing a smartphone by referring to information such as price, color or year as an evaluation criterion. As a common activation region in the choice tasks by all evaluation criteria, the bilateral occipital gyri had significant activation. The result is consistent with reports of previous studies which indicate that the occipital gyrus is the brain region related to visual processing and preference-rated tasks. Conversely, with specific activation regions involving color choice tasks, the left fusiform gyrus, left insula and right precuneus are significantly activated. The result suggests that attention to color choice becomes greater than other choices. Finally, we investigated the representation of evaluation criterion categories in decision-making using the multi-voxel pattern analysis (MVPA) method. Several studies using traditional analysis have attempted to explain the neural mechanisms associated with decision-making based on abstract rewards. However, brain-decoding research that utilizes the MVPA, especially research focusing on decision-making, remains limited. In brain analysis, decoding

strategies for multivoxels are required for various decision-making evaluation criteria. This is because, in daily life, the human decision-making process makes use of many evaluation criteria. In the present study, we investigated the representation of evaluation criterion categories in decision-making using functional magnetic resonance imaging and MVPA. Participants performed a decision-making task that involved choosing a smartphone by referring to four types of evaluation criteria. The regions of interest (ROIs) were the ventromedial prefrontal cortex (vmPFC), nucleus accumbens (NAcc), and insula. Each combination of the four evaluation criteria was analyzed based on a binary classification using MVPA. From the binary classification accuracy obtained from MVPA, the regions that reflected differences in the evaluation criteria among the ROIs were evaluated. The results of the binary classification in the vmPFC and NAcc indicated that these regions can express evaluation criteria in decision-making processes.

Acknowledgements

I am deeply grateful to my supervisor, Associate Professor Yukinobu Hoshino, for his unremitting guidance throughout the nine years leading up to my degree application. I would like to thank my co-supervisors, Professor Xiangshi Ren and Associate Professor Hiroshi Kadota, for their excellent observations, ideas, and creativity that inspired my dissertation. I would also like to thank the rest members of my thesis committee, Professor Hiroaki Shigemasu and Professor Kyoko Shibata, for serving as my committee members even in hardship. I would like to offer my special thanks to Dr. Cao Thang at the MITECH Corporation and Associate Professor Upaka Rathnayake at the Sri Lanka Institute of Information Technology for their constructive advice and interesting ideas. Special thanks also to Dr. Yuki Shinomiya, Dr. Dang Tuan Linh, Mr. Namal Rathnayake, and other members in Hoshino Laboratory at Kochi University of Technology for creating the best working conditions. In addition, I would like to thank the International Relations Center staff at Kochi University of Technology for supporting me. Finally, I would like to show my greatest appreciation to my family, especially, my wife and my son, for giving me their continuous support and understanding.

Contents

Abstract i						
Ac	knov	wledgements	vii			
1	Intro	oduction	1			
	1.1	Motivation and Overview	1			
	1.2	Structure of the Dissertation	3			
2	A Ti	ransition of Brain Activation in Problem-Solving by Long-Term Learn-				
	ing		5			
	2.1	Introduction	5			
	2.2	The Tower of Hanoi	7			
	2.3	Experimental Work	10			
	2.4	The Long-Term Learning Experiment Environment	13			
	2.5	Results	16			
	2.6	Discussion	19			
3	Con	nmonality and Specificity of Brain Activity Patterns Represented by the				
	Eval	luation Criteria	25			
	3.1	Introduction	25			
	3.2	Materials and Methods	29			
		3.2.1 Participants	29			
		3.2.2 Task and Stimuli	29			
		3.2.3 MRI Acquisition and Data Preprocessing	32			

	3.2.4	fMRI Analyses	32
	3.2.5	Multi-Voxel Pattern Analysis	33
3.3	Result	S	34
	3.3.1	Whole-Brain Analysis Results: Common and Specific Brain	
		Activity Regions	34
	3.3.2	ROI-Analysis Results: Brain Activity in the vmPFC	37
	3.3.3	Multi-voxel Pattern Analysis Results: Brain Regions Involved	
		in Decision-Making	39
3.4	Discus	ssion	43
	3.4.1	Common and Specific Brain Regions Related to Decision-Making	43
	3.4.2	Brain activation patterns in the decision-making	45

4 Conclusion

49

List of Figures

2.1	The ToH experimental task.	8
2.2	Controller used in the experiment while in the fMRI machine	10
2.3	Mirror screen in the fMRI machine	11
2.4	How to control the disks in the ToH	11
2.5	Task sequence during one session.	12
2.6	Experimental flow over 7 weeks.	14
2.7	The evaluation method for the degree of progress	17
2.8	Learning curve for the degree of progress	18
2.9	Trends in brain activities in different locations of the brain. For all	
	plots, the $Mean \pm SE$ is displayed. Asterisks indicate <i>P</i> -values (n.s.	
	$P > 0.05, *P \le 0.05, **P \le 0.01, ***P \le 0.001$ for the two-tailed	
	Welch's t-test).	22
3.1	Experimental timeline.	31
3.2	Brain regions with stronger activation in response to price vs. dummy,	
	color vs. dummy, and year vs. dummy.	36
3.3	Activation degree during each choice, left: right occipital gyrus, right:	
	left occipital gyrus	37
3.4	Brain regions with stronger activation in response to color vs. price	
	and year	37
3.5	Brain regions with strong activation on the vmPFC	39
3.6	Correct answer rate in the vmPFC	40
37	Correct answer rate in the insula	41

5.6 Correct answer rate in the NACC		3.8	Correct answer rate in the NAcc .		•		•				•					•	•			•						41
-------------------------------------	--	-----	-----------------------------------	--	---	--	---	--	--	--	---	--	--	--	--	---	---	--	--	---	--	--	--	--	--	----

List of Tables

2.1	Peak coordinates with the most differences in activation	21
3.1	List of evaluation criterion contents	31
3.2	Common significant brain activation in each contrast	35
3.3	Specific significant brain activation in color vs. (price & year)	37
3.4	Significant brain activation on the vmPFC during price, color, and	
	year choice	39
3.5	Average accuracies and standard errors by binary classification be-	
	tween each evaluation criterion	42

Chapter 1

Introduction

1.1 Motivation and Overview

In human life, people make decisions hundreds to thousands of times a day. Life involves many decisions, ranging from mundane to critical. These decisions are based on life experiences and values and lead to personal growth and goal achievement. People make decisions all the time throughout their lives, and these decisions shape their experiences and help them achieve their goals. These decisions range from simple day-to-day choices to more important long-term decisions that greatly impact a person's life. Factors such as values, past experiences, and personal goals all shape the decisions people make.

This dissertation focuses on two things related to decision-making. One is a transition of brain Activation in decision-making at problem-solving by long-term learning. We focus on learning executive functions, which include decision-making, cognitive processing, and planning. Regarding learning, memories and expressions are established by biological feedback, such as visual perception and haptics. This feedback affects the learning processing time and efficiency. The executive function consists of the same lower-level functions, such as divided attention, processing ability for multiple tasks, conversion ability of a thought set, thinking speed,

and inductive guessing [1]. Impaired executive function significantly influences individual ability, planning, prioritization, organization, attention and detailed memory, emotional response control, and planning and decision-making [2], is considered to include the frontal lobe. Executive function and planning require foresight and empirical knowledge. Planning and decision-making become logical choices from available options. In other words, this can explain the evaluation and selection of several conflicting alternatives. When making a good choice, it is necessary to measure the advantages and disadvantages of all considerations. For effective planning and decision-making, predicting each option's outcome from a task operation are essential. The best item from all options depends on the outcome predicted for the situation. Therefore, all planning and decision-making result from a definitive choice [3]. From a recognition perspective, interactions in a task environment are integrated into ongoing procedures/proceedings and are significant in the planning and decision-making process. The planning and decision-making results are related to an unchangeable selection based on logic and rationality.

The other focus point is brain representation by human choice in purchase decisionmaking. We focus on evaluation criteria, which is the motivation for choosing decisionmaking. The field of neuro-marketing, which explores the human decision-making process by objectively measuring brain reactions and taking advantage of brain science and psychophysics knowledge, is attracting increasing attention [4, 5]. The neuroscience field uses fMRI, functional near-infrared spectroscopy, positron emission tomography, and electroencephalography to explore human decision-making processes[6, 7]. Several studies using traditional univariate analysis have attempted to elucidate the neural mechanisms associated with decision-making based on abstract rewards.

Multi-voxel pattern analysis (MVPA) has been attracting increasing attention, as shown in recent fMRI studies to elucidate brain activity patterns [8–10]. A relatively

large amount of the existing research using MVPA focuses on decision-making because it is important to understand which brain regions are involved in the various metrics used during decision-making in daily life. Given this background, the present study aimed to estimate user choice in decision-making based on brain activity.

1.2 Structure of the Dissertation

This dissertation consists of four chapters. The content of the dissertation is as follows.

- Chapter 1 introduces the motivation and overview of this research.
- Chapter 2 focuses on brain activity before and after learning. The effect of the planning and decision-making process concerning executive function on brain activity via long-time learning was investigated.
- Chapter 3 focuses on brain activation during the decision-making process, which was selected based on a single evaluation criterion. The fMRI result showed brain regions with common brain activation and specific brain activation during the selection task. Moreover, MVPA results showed that some brain regions could express the influence of the evaluation criteria during decisionmaking.
- Chapter 4 concludes this dissertation.

Chapter 2

A Transition of Brain Activation in Problem-Solving by Long-Term Learning

2.1 Introduction

Identifying the areas of significant activation in the brain and their relation to executive function, including cognition, as a particular brain mechanism [11], is essential to clarify the effects to investigate the impact of planning and decision-making processes in long-time learning. Executive function is defined as a goal–direction behaviour, including planning for the short-term future. The ability to maintain an appropriate attitude for achieving future goals relies on four main points, according to Lezak et al., such as goal formulation, planning, carrying out goal-directed plans, and effective performance [1]. Executive function involves higher-order information from any sensory stimulus. Higher-order executive function processes transmit information to the brain. The prefrontal cortex (PFC) is closely involved in executive function and plays crucial roles in planning, executive processing, and emotional expression. The ToH task measures planning abilities by systematically varying planning demands [12–14].

In the ToH task, a player must rearrange a set of multiple disks on three pegs

6

of varying sizes from the start state to the goal state using the minimum number of moves. Completing this task is known to be critically dependent on PFC activation [11, 15–22]. Therefore, the ToH task is considered appropriate for investigating the role of the PFC.

On the other hand, previous studies have confirmed activation of the dorsolateral PFC (DLPFC) during the ToH task. Some researchers have argued that the ToH task activates the DLPFC (BA 9), parietal lobe (BA 7, 40), inferior frontal gyrus (BA 44), and premotor cortex (BA 6) [23–27].

Card et al. studied the minimum reaction time when starting an exercise after a short-time judgment of human acquisition from visual information. The model human processor (MHP) is a famous model of human processing developed by Card [28]. Humans have a minimal response time of more than 370 ms (eye movement processor = 230 ms, cognitive processor = 70 ms, motor processor = 70 ms) from the time information is sent and the planning and decision behavior. Puzzle tasks in this experiment do not have any delay or prediction error between the screen information and behavior. By pressing a button, participants can predict the next state based on planning and decision-making.

Human decision-making is a complex process involving many factors. There may also be a stochastic element in it. However, human decision-making is generally regarded as a decisive process, and stochastic factors are often not considered. Also, Human decision-making is a complex process that involves many factors, such as cognitive biases, emotions, past experiences, and social influences. It may also involve a degree of randomness or stochasticity, as the individual may not have all the information needed to make a fully informed decision. However, it is generally viewed as a decisive process, and the role of stochastic factors may not be fully considered. It can be said that operators can make good decisions when they have enough information. In this case, having accurate and relevant information is necessary to make good decisions. In the ToH task, participants have the whole information for a solution, there is no stochastic element. The operation depends on precise planning and decision, and participants can perform the task in less time. The time lags during these operations are caused mainly by the planning and decision-making process based on visual information. In some control cases for various planning and decision-making processes, the time lag is about 400–500 ms [28]. If learning levels are sufficiently advanced; participants can control these times. Trends in the time lags in button pressing correlate with trends in the temporal learning time. fMRI and outside training environments have the same time lag and learning properties.

The present study focuses on the brain regions mainly involved in executive function, including the DLPFC, parietal lobe, inferior frontal gyrus, and premotor cortex [23–27]. Our investigation is to observe changes in the degree of activation depending on the learning status of the learners.

This long-time learning experiment was conducted using a ToH puzzle, frequently used to measure learning ability [11, 29–32]. The long-time learning experiment consisted of a learning term and three MRI terms. In the learning term, learners studied the ToH puzzle outside the fMRI scanner for seven weeks. The three MRI terms were conducted at the end of the learning term's starting, middle, and final third.

2.2 The Tower of Hanoi

The ToH is a popular puzzle in cognitive psychology and neuropsychology used to assess a set of behaviors collectively referred to as executive function. In addition, the ToH is also a popular puzzle game for cognitive science and neuroscience [11, 29–32]. This task is typically used to evaluate behaviors and executive function. In the current experiment, estimating the brain activation recursively was attempted. Inductive inference is an estimation method that identifies general rules of individual, partial, and special events [33, 34]. The ToH puzzle is a complex cognitive task in which participants must learn the procedural process for disk operation. The guidelines of this game can be explained as follows.

The detailed rules are as follows:

8

- 1. The ToH puzzle consists of three poles and different-sized disks.
- 2. In the initial position, all disks are stacked on the left pole in ascending order.
- 3. Participants can only move one disk at a time.
- 4. The disks can be moved from one pole to another.
- 5. The disks can be placed on an empty pole or larger disks.
- 6. A larger disk cannot be placed on top of a smaller disk.
- 7. The goal is to move all the disks to the right pole, as shown in Figure 2.1.
- 8. For *n* disks, the optimal solution path is 2^{n-1} moves.
- 9. The game ends when all disks are moved to the right pole.



FIGURE 2.1: The ToH experimental task.

The new hypothesis consists of the procedure of trial and error. Players acquire particular strategies by trial and error method in the ToH puzzle. These trials contribute to the final solution to reproduce the completed moving procedure. The optimal number of movements for the three disks in the ToH puzzle is seven; however, if the frequency of the trial and error decreases, the thinking process regarding executive function shifts to a working memory task instead. This process depends on the ability of the executive function.

The planning ability is enhanced by the trial and error method. In this case, the planning time would be longer than that for working memory. The brain regions associated with executive function activate if a player is making plans for a long time.

After that, this process would shift to a working memory task. Due to the complexity of the task, players cannot memorize the whole process. Players can memorize patchy rules of the solving process. Hence, those disjointed rules act in widearea memory regions.

In the main experiment, participants learned the ToH task for seven weeks. The frequencies of button presses and achievements were recorded for each participant. Difficulty levels can be modified according to game times and the number of disks. In our analysis, the repeat task was prepared as a control task. In repeat tasks, participants push a button and do not play the game. Brain activities during the repeat task are independent of planning and decision-making. At the same time, the participant watches a video of a previous ToH task. This task has the same visual stimulus as the ToH task.

Moreover, participants perform these tasks without using executive functions. Thus, this repeat task is performed only by pushing a button and not playing the ToH. It is a type of dummy operation. In this case, particular responses appear in the motor and visual cortices. This analysis method uses two types of fMRI scans.

The first one is the scan of the ToH task. These data involve the use of the executive function. The other scans the repeat task image data without the executive function. Calculating differences between tasks and repeats can check the activation of executive functions.

2.3 Experimental Work

Before the experiment, the participants were interviewed about their knowledge and experiences with the ToH puzzle. All participants had some basic knowledge about the ToH puzzle. Player knowledge of the ToH was the same for the experimental baseline. Therefore, this knowledge can be defined as a social composition condition.

A screen located in front of the participant showed the ToH task. The participant solves a five-disk ToH puzzle in the experiment. The task is to play the ToH using a four-button controller, as shown in Figure 2.2. Participants can attempt to solve the puzzle on a screen using a controller, as shown in Figure 2.3. The participant lies inside the fMRI to play the ToH puzzle task while holding a controller in the right hand. The controller has four buttons, and this experiment uses the left, top, and right buttons corresponding to the left, center, and right poles in the ToH.



FIGURE 2.2: Controller used in the experiment while in the fMRI machine.



FIGURE 2.3: Mirror screen in the fMRI machine.

Participants can press a button corresponding to a pole on which disks are stacked. When the button is pressed, the disk at the corresponding pole is lifted, as shown on the left in Figure 2.4. Next, the participant can press a button corresponding to a target pole. The lifted disk moves to the target pole at that time, as shown on the right in Figure 2.4. The task is considered complete after all disks are on the right pole. Then, the disks are reset to the left pole as the initial position for the next game. Participants repeated the ToH task continuously within a given time.



FIGURE 2.4: How to control the disks in the ToH.

Figure 2.5 shows one session of the MRI experimental sequence. This experiment consisted of three tasks and three repeats. In the Task 1 period, the task started from the initial state. The purpose was to confirm the activation associated with executive function based on activation differences between task and repeat scans. All disks were set on the left pole as the initial position. The disk starting position in Task 2 was from the ending position in Task 1. The disk starting position in Task 3 was from the ending position in Task 2. Tasks 1, 2, and 3 only differed regarding the disk starting position. It repeated 1, 2, and 3, and a video of Tasks 1, 2, and 3 was shown. All task and repeat times were 40 s. The rest time between tasks was 10 s, and the rest time between tasks and repeats was 15 s. The total time of one session was 5 min and 40 s (136 images were scanned).



FIGURE 2.5: Task sequence during one session.

During the task period, the participant played the ToH. Participants continuously played as many times as instructed. During the task 1 period, the task started from the initial state. All disks were set on the left pole as the initial position. During the Task 2 period, the participants continued solving the puzzle from the final state of the previous task.

During the repeat period, the participants synchronously pressed the button while watching the video recorded during the task period. During this period, the push button did not affect the time performance or executive function tasks. Therefore, images during this period included only activation information in the motor and visual cortices based on controller operations. They did not have brain activation information regarding executive function, and the experiment was intended to extract only activation information. This image differed between the task and repeat blocks. During the rest period, a fixation cross was displayed for the participant to focus. The color of the fixation cross depended on the next period. If the task period was next, the fixation cross color was red. If the next was the repeat period, then the color of the fixation cross was light blue.

In data analysis, realign (motion correction), normalize (standardization), smooth (smoothing), specify-1st-level-modeling (standard brain model determination), and estimate (brain activity grayscale images are created in the flow of the estimated activity region). By taking the difference between them, the active part is determined. Active areas are narrowed down to p < 0.001 significant active areas by *t*-test.

2.4 The Long-Term Learning Experiment Environment

The long-time learning experiment was conducted with four participants. These experiments need a longer time to process the scanning of 402 image data samples. The scanning processes cost a lot of money, and a lot of time. Therefore, finding volunteers for these types of studies is difficult. Considering all of these concerns, only four samples were tested for this experiment. However, to have a uniform study, the participants were chosen at similar ages (21–23 years old). In addition, both males and females were considered for these experiments. The attributes of the participants are as follows.

- Participant 1 was male, 21 years old, and right-handed.
- Participant 2 was male, 23 years old, and right-handed.
- Participant 3 was female, 21 years old, and right-handed.
- Participant 4 was male, 21 years old, and right-handed.

This study was approved by the Research Ethics Review Committee of Kochi University of Technology (approval number: 52-C3). All participants provided written informed consent before the experiment began. Figure 2.6 shows the experimental flow. This learning experiment was conducted for seven weeks. The participants learned to solve the ToH for 40 min once or twice a week outside the MRI. Participants 1 and 2 participated in 12 learning sessions (L1–L12 in Figure 2.6). Participants 3 and 4 participated in a total of 10 learning sessions. Though the number of sessions used to train the participants differed (10 and 12), the evaluation was done constantly. Therefore, overlearning by participants 1 and 2 can be disregarded.

To observe the participants' brain activity and progress in and convergence of learning, they underwent fMRI scans three times while solving the ToH (M1–M3 in Figure 2.6) in this experiment. fMRI scans were performed immediately after the first (L1), fourth (L4), and last learning periods (L12 or L10). The interval between M1–M2 was ten days, and the interval between M2–M3 was 30 days.



FIGURE 2.6: Experimental flow over 7 weeks.

It was used a MAGNETOM Verio 3T scanner (SIEMENS Co., Ltd., Erlangen, Germany) for acquiring fMRI images. Head movement was limited with the use of mild restraints and cushioning. The imaging parameters were TR = 2500 ms, TE = 30 ms, $FoV = 192 \text{ mm}^2$, voxel size = 3.0 mm^3 , and slice thickness = 3.0 mm. 136 scanned for analysis images per session. Two images were excluded from the analysis because the longitudinal magnetization of the tissue was unsteady.

Furthermore, an anatomical image with a resolution of 1.0 mm³ was combined with a T1-weighted image to obtain positional information. A projector was installed outside the MRI room and projected the experimental task image on a resin screen near the head-side opening of the fMRI device through a telephoto lens. The participants watched the images on the screen via a mirror placed over the coil above their heads.

In this experiment, the participants performed a ToH task involving five disks. Participants lying on the bed inside the fMRI machine performed the ToH task using a controller in their right hand. Participants repeatedly carried out the ToH task within a given time. The experimental design employed was a block design with alternating tasks and rest periods. Performance data during the experiment were recorded to confirm the learning progress.

Moreover, fMRI images were obtained for each learning process. The five-disk ToH task was performed inside the fMRI machine, and the time allotted for each task was 40 s. Three ToH tasks using five disks are blocks within a certain period.

Software Platform of Experiments

SPM12 software (Wellcome Centre for Human Neuroimaging, London, UK) was used to process and analyze the fMRI data. A total of 408 images (136 images for each M1, M2, and M3 instance) were obtained from three fMRI runs. The first two scan images were discarded from each fMRI- run. Thus, this analysis used 402 scan images ($408 - (2 \times 3)$ instances).

These first two scan images were discarded because the magnetization of the MRI was not in a steady state at the beginning of each scan. Functional images were corrected for differences in slice acquisition time and motion artifacts.

This analysis examined the degree of activation between the early and late stages of learning at each region of interest (ROI). The data were realigned, normalized according to the standard Montreal Neurological Institute (MNI) model, and smoothed with an 8-mm full-width–half-width Gaussian filter. MNI coordinates were used for the brain activation analysis. Only focused voxels were analyzed using the WFU PickAtlas toolbox for masking [35, 36].

2.5 Results

The degree of progress was the learning level indicator for all participants. The calculation method for the degree of progress is as follows.

The first step was to find the shortest number of moves remaining on all the boards. The minimum number of moves for the 5 disks was 31 from the initial position, and the minimum number of remaining movements is less than 31 moves from the other starting position. One point was added to the degree of progress when the remaining operations decreased by one step. If there was no change in the remaining operation, the degree of improvement remained the same. Conversely, one point was subtracted from the degree of progress when the remaining operations increased by one step. Therefore, the total possible score when solving a puzzle was 31 points. The remaining operation was assumed to be a measurement of the learning level. The point calculations are shown in Figure 2.7.

Card's MHP defines the fastest time a person can press the button as approximately 0.3 s [28]. In this case, since the time for one task, was 40 s, the maximum number of evaluation points was about 130 (=40/0.3). If there are few remaining operations in the task, the degree of progress is a high score. In this case, a high score indicates a quick and accurate operation. Therefore, in this case, the learning level of the participant is high. The degree of progress is calculated to evaluate the



learning level for the task.

FIGURE 2.7: The evaluation method for the degree of progress.

The participants completed the tasks between 150 and 180 times and underwent fMRI scans nine times. The degree of progress was calculated for each job. Figure 2.8 shows the learning curves for the degree of improvement for each participant. Their transitions can be confirmed based on the learning curves.

The learning curves converged after Task 75 (in L5). The gray part shows corresponding marks about the data from the fMRI scans (M1–M3). The observed learning progress for M1 and M3 are at the beginning and end areas of the learning period, respectively. Moreover, their learning was progressing in M2.

Four ROI masks were used for the DLPFC, parietal lobe, inferior frontal gyrus, and premotor areas. Several examples show an increasing or decreasing trend in brain activity as learning progresses [37, 38]. There is a close relation to the learning behavior of brain regions that show such increasing/decreasing trends. In the first analysis, we observed differences in brain activity between the whole brain's first and third fMRI scans. We searched for brain regions (voxels) that showed significant differences in differences in brain activity level of each scan for the regions that showed significant differences in brain activity. We designed the contrasts and analyzed the brain activity on the SPM to verify this significant difference. The contrasts were "first fMRI scan vs. third



FIGURE 2.8: Learning curve for the degree of progress.

fMRI scan" and "third fMRI scan vs. first fMRI scan". These fMRI scans consisted of all tasks vs. all repeats in the same session. Statistical thresholds were set for uncorrected brain peak levels (p < 0.001) and corrected cluster levels (p < 0.05). For motion correction, all models included the six-dimension head-motion parameters as the regressor.

Executive function is critically dependent on PFC activation [11, 15–22]. The PFC is believed to be involved in planning ability [2, 39, 40].

2.6 Discussion

Previous studies have confirmed the activation of the DLPFC (BA 9) in the ToH task. On the other hand, some researchers have argued for the activation of the parietal lobe (BA 7, 40), inferior frontal gyrus (BA 44), and premotor cortex (BA 6) [23–27]. These studies suggest that these brain regions are involved in planning ability. Investigations of these brain regions focus on the transition of brain activity at the peak coordinates in each task.

Moreover, investigations of differences in activation have focused on these regions in each participant. Table 2.1 shows the location of the brain regions with the maximum difference in activation between the first (M1) and third fMRI scans (M3) for each participant. Figure 2.9a–d shows the degree of brain activity for each of the three tasks (M1, M2, and M3) in the experimental flow. The reports of brain activation were indicated with signal plots for the DLPFC, premotor cortex, parietal lobe, and inferior frontal gyrus. All signal plots show activity under the experimental condition (event type) relative to baseline (in arbitrary units [a.u.], \pm 90% confidence interval). The plots show activity patterns at the peak of activation (i.e., single voxel) as selected from the whole-brain contrast SPM map.

Performance data showed that each participant's learning curve converged at the 100th and 150th tasks. The goal of this task was to observe only the brain activity

related to executive function in each learning period. Reports have shown that executive function is closely associated with the parietal lobe and cerebellum, especially the PFC. The PFC has involved the establishment of target behaviors that are necessary for executive function. Environmental dependence is also said to be involved in goal maintenance. Moreover, conservation is involved in the flexible changing of goals.

Region label	Participant No.	L/R	MNI coordinates						
			x	у	z				
Parietal lobe	1	L	-22	-10	72				
	2	L	-4	6	66				
	3	R	38	10	60				
	4	L	-22	-2	66				
Inferior frontal lobe	1	R	44	-70	48				
	2	R	58	-24	14				
	3	L	-26	-66	52				
	4	L	-30	-50	62				
Premotor cortex	1	L	-32	34	36				
	2	L	-8	26	34				
	3	L	-40	46	32				
	4	L	-44	32	34				
Dorsolateral prefrontal cortex	1	R	58	16	18				
	2	L	-60	16	12				
	3	R	46	16	14				
	4	R	52	12	8				

TABLE 2.1: Peak coordinates with the most differences in activation

Based on the results of the fMRI analysis, differences in activation were observed in the DLPFC (BA 9), parietal lobe (BA 7, 40), inferior frontal gyrus (BA 44), and premotor cortex (BA 6) in each participant. Comparing the brain activities during M1– M3, the differences followed decreasing trends. These brain regions are involved in



(a)Trends in brain activity in the premotor cortex.



(b)Trends in brain activity in the parietal lobe.



(c)Trends in brain activity in the dorsolateral prefrontal cortex.



(d)Trends in brain activity in the inferior frontal gyrus.

FIGURE 2.9: Trends in brain activities in different locations of the brain. For all plots, the *Mean* \pm *SE* is displayed. Asterisks indicate *P*-values (n.s. *P* > 0.05, **P* \leq 0.05, ***P* \leq 0.01, ****P* \leq 0.001 for the two-tailed Welch's t-test).
the executive functions of complex behaviors, goal maintenance, flexible goal modification, and a combination of goals [2, 23–27, 39, 40]. There have also been reports of decreases in related brain region activity as learning progresses [38, 41, 42].

Furthermore, the performance results in this experiment confirmed a convergence of the learning curve. We believe that the participants were in a state where they had completed sufficient learning and could perform efficiently.

In the fMRI data, brain regions involved in executive functions showed a decrease in brain activity after learning relative to before learning. This result suggests that the participants efficiently performed task anticipation, planning, and decisionmaking.

On the other hand, several other studies employing the ToH task have reported that the frontal pole is related to executive function [19, 29]. However, the results of this experiment, which measured differences between pre-and post-learning, found no differences in the activation of this region.

This result suggests that there is not much effect on short-term learning for the frontal pole. Hence, long-time learning performed with trial and error increases brain activity.

Chapter 3

Commonality and Specificity of Brain Activity Patterns Represented by the Evaluation Criteria

3.1 Introduction

Human decision-making, both simple and complex, occurs hundreds to thousands of times per day. A few examples may include decisions such as "What to eat", "Which clothes to wear", and "How to solve a problem". Among these scenarios, purchase decision-making is a familiar and frequently encountered decision. When buying a product, especially a more expensive one, people tend to compare information to aid in the decision-making process. In recent years, online shopping has given people the opportunity to purchase a greater variety of products. However, due to the large variety of product lineups, it is difficult to make the best decision that satisfies consumer requirements. Furthermore, consumers cannot check the actual products, only view the product information, images, and reviews displayed on the screen. Therefore, decision-making regarding the purchasing of products should be supported. An important step in this direction would be to elucidate brain representation to gain a better understanding of the human decision-making process.

One area that has been repeatedly shown to be activated by diverse rewarding stimuli is the vmPFC. In several neuroimaging studies, this brain region is active for a variety of primary and abstract rewards, including sports cars [43], cola preferences [44], pleasant odors [45], wine prices [46], facial attractiveness [47] and money [48]. The results of these past studies support the idea that the vmPFC is involved in converting the values of diverse stimuli into a common measure for a behavioral choice. Other findings have suggested that product preference activates the nucleus accumbens (NAcc). Knutson et al. [49] reported that preference elicits NAcc activation before a purchasing decision, whereas excessive prices can elicit insula activation and medial prefrontal cortex (MPFC) deactivation.

Multi-voxel pattern analysis (MVPA) considers external stimuli, motion state, and mental content to be encoded in brain activity patterns. MVPA can save and distinguish spatial response patterns lost by averaging the responses across voxels in the ROI, as in univariate analysis. When detecting the presence of a specific cognitive condition in the brain, the main advantage of MVPA is its increased sensitivity. The conventional fMRI analysis attempts to find voxels that exhibit statistically significant responses to experimental conditions. To increase the sensitivity to certain conditions, these methods spatially average the voxels that respond significantly to that condition. Although this technique reduces noise, it also reduces the signal in two important ways. First, a voxel with a non-significant response to a particular condition can carry some information about the presence/absence of that state. Second, spatial averaging can blur the spatial pattern that distinguishes experimental conditions. As with conventional methods, the MVPA approach also attempts to increase sensitivity by looking at the contributions of multiple voxels. However, to avoid the signal loss problem mentioned above, MVPA does not routinely involve a uniform spatial average of the voxel response. Instead, it uses a weighted average of responses that treat each voxel as a separate source of information about the participant's cognitive state. This technique optimizes these weights and aggregates this information among voxels and finds ways to guide participants more accurately in terms of what they are thinking [50]. With this method, by analyzing a pattern composed of multiple voxels, it was possible to explore the brain expression of detailed information that could not be examined by conventional fMRI analysis. The traditional univariate approach focuses on the activity changes of each voxel. By contrast, MVPA extracts information from many brain locations (voxels) at the same time to examine the spatial brain activation pattern. MVPA is often used for neural decoding. Neural decoding is a technique to estimate stimuli, behaviors, and cognitive states. Several neural decoding studies are being conducted to reconstruct visual information, cognitive judgments, and emotions [51-53]. Machine learning methods such as the support vector machine and neural network are important for this approach. Neural decoding is realized by learning brain activity patterns, which are multidimensional variables, using a machine learning algorithm and outputting prediction values from new brain activity patterns using the learned model. There are many studies on biometric data classification using machine learning other than neural decoding. In the medical field, research is being conducted to detect lesions and classify benign/malignant tumours, and machine learning methods are used [54, 55].

Given this background, the present study aimed to estimate user choice in decisionmaking based on brain activity. To achieve this aim, we experimented to verify the brain regions involved in the evaluation criteria in decision-making processes and investigated the representation of their categories using fMRI and MVPA. Assuming a situation involving the purchase of smartphones through online shopping, participants picked one from a choice of two products with information for a single evaluation criterion. This task involved four types of evaluation criteria for decisionmaking. We focused on the voxel pattern in ROIs based on differences in the type of evaluation criteria. To our knowledge, brain regions reflecting the type of evaluation criteria used in decision-making have not been reported.

We hypothesized that the type of evaluation criteria is specific to or commonly involved in certain brain regions. This analysis focused on the vmPFC, insula, and NAcc, which are considered to be involved in decision-making and were used as ROIs[56–60]. We hypothesized that there would be a difference in the activation patterns of the vmPFC, insula, and NAcc depending on the type of evaluation criteria. To examine this hypothesis, a decision-making task regarding each evaluation criterion was performed. Differences in brain activation patterns due to the differences in evaluation criteria in these ROIs were then examined by comparing each combination of evaluation criteria. For the analysis, we used MVPA, which has been established as an effective method for identifying and classifying brain activity patterns. Based on different evaluation criteria, the voxel patterns from all associated regions were examined using MVPA. Each combination of the four evaluation criteria was analyzed after binary classification by MVPA. Finally, based on the binary classification accuracy obtained from MVPA, the regions among the ROIs that reflected differences in evaluation criteria were evaluated.

For many people, evaluation criteria such as price, color, and production date are important factors in making a better purchase. The results obtained in this experiment indicate that vmPFC and NAcc respond concerning the same endpoints when making a choice. Moreover, as shown in [46, 49, 56, 57, 59], the vmPFC and NAcc are sites involved in decision-making and execution. On the other hand, there are neuromarketing methods that take advantage of findings from brain science and apply them to marketing activities [61–63]. It is important to analyze consumer psychology and behavioural principles indicated by brain science. The traditional methods of behaviour analysis were questionnaires or interviews and thus could not elicit consumers' unconscious true intentions. However, the results of this research proved that the vmPFC and NAcc measurements can be used as a model to visualize and quantitatively evaluate consumers' unconscious psychology and preferences, which are difficult to verbalize.

3.2 Materials and Methods

3.2.1 Participants

Twenty-five participants (five females; two left-handed; mean age 20.60 years, standard deviation 1.26 years, age range 19–22 years) participated in the fMRI experiment. One participant who did not complete the experiment was excluded. Therefore, 24 participants were finally included in the data analysis. This study was approved by the Research Ethics Review Committee of Kochi University of Technology (approval number: 52-C3). All participants provided written informed consent before the experiment began.

3.2.2 Task and Stimuli

All participants performed a decision-making task that involved choosing a smartphone by referring to each evaluation criterion. Fig 3.1 shows the experimental timeline. This experiment is based on a block design. It consists of several discrete periods of on–off blocks, with the "on" representing a task condition, and the "off" referring to a rest state or different task condition.

In this task, participants viewed a screen, an example of which is illustrated in the lower part of Fig 3.1. The screen presented two identical smartphone images and the letters of each different combination of evaluation criteria as stimuli. This experimental design using these stimuli, which are illustrated in two identical images, each with different information, has also been adopted in other decision-making studies [64]. The participants pushed the left or right button to select the smartphone they wanted more.

The experimental conditions included the price, body color, and production year as the evaluation criteria. The body color evaluation was conveyed through textual information instead of an illustration. The reason for this was to avoid the possibility of a difference in willingness to purchase depending on the displayed color of the product [65]. These choices were selected as evaluation criteria because they can be evaluated easily regardless of the presence or absence of smartphone knowledge among the participants. In addition, a dummy variable (four squares as meaningless symbols) was set as the control condition. Brain activity at the time of the main task was considered to include three main effects: decision-making, visual recognition of the stimuli, and button-pressing at the time of selection. In addition, brain activity during dummy tasks is thought to reflect the effects of almost the same condition as the main task, except for decision-making. Brain activity regarding the difference between the main and dummy tasks is considered to represent only the effects of decision-making. Each criterion had four types of content. Table 3.1 shows the list of evaluation criterion labels used in this experiment. Each participant was considered to have a different priority for each evaluation criterion. In this experiment, it was hypothesized a specific brain-related region would refer to the impression for each evaluation criterion that included priority differences. The evaluation criteria appeared in a different order for each participant. Each participant performed two runs under the same conditions. Each run contained eight blocks of four separate tasks: the price choice, the color choice, the year choice, and the dummy choice. During each choice, a screen showed information for 3 seconds followed by a rest period for 2–4 seconds. The screen presented two images of the smartphone, shown on the left and right of the screen. Different labels under the images represented each evaluation criterion. In the choice tasks, the participants decided on an object

in their mind, and then pushed the left or right button to select the object they had chosen. In the dummy tasks, the participants were required to push either button intuitively. The total time of one run was 306 seconds. Stimuli images and words were rear-projected onto a screen placed in the scanner bore using an LCD projector. The screen showed two identical smartphone images and two different labels as an evaluation criterion, as demonstrated in Fig 3.1.

Evaluation criteria		Lab	els	
Price	¥10,000	¥15,000	¥20,000	¥25,000
Color	Black	White	Red	Blue

2008

2012

2016

2004

TABLE 3.1: List of evaluation criterion contents

Note: ¥ sign means Japanese yen.

Year Dummy



FIGURE 3.1: Experimental timeline.

3.2.3 MRI Acquisition and Data Preprocessing

Scanning was performed on a 3.0-tesla scanner (MAGNETOM Verio, Siemens Healthinners, Erlangen, Germany) using a 16-channel head coil at the Kochi University of Technology. Functional scans were acquired with a standard gradient-echo echoplanar imaging sequence to cover the whole brain (field of view = 192 mm^2 ; repetition time = 3,000 ms; echo time = 30 ms; flip angle = 90 degrees; slice thickness = 3.0 mm; voxel size = 3.0 mm^3). Each run of the functional scans obtained 102 volumes over a total duration of 306 seconds. A high-resolution T1-weighted anatomical scan was acquired for each subject (1.0 mm³ resolution).

The first two scans (6 seconds) in each run were discarded to account for any instability with the fMRI scanner. SPM12 software (Wellcome Centre for Human Neuroimaging, London, UK) was used to process and analyze the functional data. Functional images were corrected for differences in slice acquisition time and motion. The data were then realigned and normalized to the Montreal Neurological Institute (MNI) standard brain model. The brain activation degrees were analyzed on the MNI coordinates.

3.2.4 fMRI Analyses

The following four conditions were modelled: price choice, color choice, year choice, and dummy choice. Common or specific brain regions were involved in each condition, and these regions were identified by creating contrasts. With the first level (single subject analysis), contrasts (price vs. dummy, color vs. dummy, and year vs. dummy) were created to identify brain regions, which were commonly activated for all contrasts. Price choice vs. the two other choices, color choice vs. the two other choices, and year choice vs. the two other choices were created to identify brain regions, which were specifically activated for each contrast. With the second level (group analysis), one sample t-tests were performed to examine significant brain activation among the group during the contrasts mentioned. A statistical parametric

map was generated using the price, color, and year vs. dummy choice contrasts. Clusters of voxels were corrected for multiple comparisons across the whole brain using family-wise error correction and a threshold of p-values: p < 0.05 [66]. The statistical parametric maps were generated using color vs. (price & year) choice contrast. Clusters were defined using a height threshold of p < 0.001 uncorrected for multiple comparisons with a cluster size threshold of k = 171 voxels. In the contrasts of price vs. (color & year) and year vs. (price & color), no suprathreshold clusters were applied.

3.2.5 Multi-Voxel Pattern Analysis

The results of the fMRI analysis described in the previous section showed activation in the vmPFC during decision-making about each evaluation criterion. Brain activation patterns in the decision-making task were investigated using MVPA based on each evaluation criterion.

MVPA was performed using a support vector machine (SVM) with a linear kernel [67], as implemented in the Pattern Recognition for Neuroimaging Toolbox [60]. The pattern analyses were performed separately for each participant. The β value at decision-making based on each evaluation criterion obtained from the general linear model in the previous section was taken as the input value. There were 48 β belonging to four evaluation criteria, including the dummy, from two runs for each participant (24/run). Binary classification according to the four evaluation criteria was carried out, with each *beta* representing a single decision-making event based on an evaluation criterion. As the evaluation criterion for each β is already known, in MVPA, these *beta* values were labelled "price", "color", "year", and "dummy". The *beta* values, which were labelled as two types of evaluation criteria, were input to the SVM as training sets to generate a boundary of two classes. We examined into which class the sample data with either label were classified. The analysis was performed using voxels in only the vmPFC, insula, or NAcc by masking using the PickAtlas toolbox. In this analysis, binary classification by SVM was performed to test whether every ROI could distinguish between the four evaluation criteria (e.g., "price or color", and "color or year"). WFU PickAtlas [35, 36] was used to create the three ROI masks for the vmPFC, insula, and NAcc. The mask for vmPFC selected the Brodmann areas 10, 14, 25, and 32 as defined by Finger [68]. Data were cross-validated using the leave one block out method, with two sets of 24 data points from each participant. Only one sample data point was extracted from all data as a test set. The remaining data points were used as a training set. The verification was repeated using all sample data as a test set. The number of correct answers among the 48 data points by each participant (e.g., when price data were classified as price) was obtained from the SVM. Next, the average correct answer rate for each binary classification obtained from all participants was calculated. The average correct answer rates were evaluated to reveal whether the brain activity pattern of each ROI expressed the evaluation criteria for decision-making.

3.3 Results

3.3.1 Whole-Brain Analysis Results: Common and Specific Brain Activity Regions

We performed a whole-brain analysis to identify regions that have common significant activation in each choice. Table 3.2 shows MNI peak coordinates in each contrast, which are price vs. dummy and color vs. dummy, year vs. dummy. MNI peaks were reported for voxels at p < 0.05 FWE corrected for multiple comparisons; L = left and R = right hemisphere. Significant differential activities in the contrast of price vs. dummy were observed in the left occipital gyrus and right calcarine. In the contrast of color vs. dummy, significant differential activities were observed in the left fusiform gyrus, bilateral occipital gyri, right lingual, bilateral superior frontal gyri, left insula, right cerebellum, left triangular part of inferior frontal gyrus, right calcarine, right middle frontal gyrus and right angular. In the contrast of year vs. dummy, significant differential activities were observed in bilateral occipital gyri, right inferior temporal gyrus, and right cerebellum.

Region Label	Cluster size	T-statistic	MN	I Coo	rdinates
			x	у	Z
Price vs. Dummy					
Occipital Mid L	482	9.35	-22	-96	0
Calcarine R	363	8.80	24	-92	2
Color vs. Dummy					
Fusiform L	484	8.94	-40	-54	-10
Occipital Inf R	59	8.41	36	-84	-12
Lingual R	91	8.01	12	-90	-4
Frontal Sup Medial R	15	7.60	6	24	42
Occipital Mid R	31	7.38	32	-66	26
Insula L	16	7.13	-30	18	-4
Cerebellum 9 R	5	6.90	6	-56	-40
Frontal Inf Tri L	4	6.83	-38	38	10
Calcarine R	7	6.83	6	-62	12
Occipital Sup R	8	6.81	30	-72	46
Frontal Sup 2 L	1	6.78	-12	50	38
Frontal Mid 2 R	1	6.75	36	52	-2
Occipital Mid L	24	6.67	-30	-78	24
Frontal Mid 2 R	1	6.53	30	54	4
Occipital Mid L	1	6.50	-28	-80	18
Angular R	1	6.44	34	-70	46
Year vs. Dummy					
Occipital Inf L	464	10.26	-38	-82	-10
Occipital Inf R	527	9.80	22	-92	-4
Occipital Mid L	23	7.69	-30	-80	24
Temporal Inf R	19	6.95	50	-64	-12
Cerebellum 6 R	1	6.45	10	-74	-18

TABLE 3.2: Common significant brain activation in each contrast

Note: Region labels were named on the basis of the automated anatomical labelling (AAL) template [69] which is a software and a digital human brain atlas with a labelled volume. The labels indicate macroscopic brain structures. Cluster size is reported in voxels. The T-statistic value is the total average, which was calculated for each voxel from the MRI data of each subject, divided by the standard deviation of all subjects.

Significant activity was observed commonly for all the contrasts only in the occipital gyrus. Figure 3.2 shows the significant activities in the occipital gyrus (coordinate: z = -12) of price choice, color choice, and year choice versus dummy choice

(*p* < 0.05 corrected).



FIGURE 3.2: Brain regions with stronger activation in response to price vs. dummy, color vs. dummy, and year vs. dummy.

The difference in price, color, and year choice on bilateral occipital gyri were investigated. Firstly, a contrast of (price & color & year) vs. dummy was set to determine the voxel (coordinate) for comparing activation degree. Then, activation degrees among the price, color, and year choice were compared on the same voxel of the determined coordinate. As a result of the analysis of this contrast, the peak coordinate of significant activation was located (32, -86, -12) and (-40, -80, -10) in bilateral occipital gyri. In these two voxels, brain activation degrees during each choice (price, color, year, and dummy) were calculated. Figure 3.3 shows activation degrees during each choice on left and right occipital gyri. The occipital gyrus is located in the visual cortex, therefore this activity was considered a possible influence by seeing the stimulus. However, brain activation degree during the price, the color, and the year choice were significantly higher than the dummy choice, this brain activation is considered to be a decision-making activity. No significant activation difference was observed among the price, the color, and the year choice.

In addition, specific significant activation in each choice was investigated. The results of color vs. (price & year) are depicted in Figure 3.4 and the peak coordinates are given in Table 3.3 (p < 0.001 uncorrected, k = 171). In this contrast, the left inferior frontal gyrus, right precuneus, and left occipital gyrus had significant activation.



FIGURE 3.3: Activation degree during each choice, left: right occipital gyrus, right: left occipital gyrus

There was no significant activation in the contrasts of price vs. (color & year) and year vs. (price & color).

TABLE 3.3:	Specific significant	brain	activation	in	color	vs.	(price	&
		year)						

Region Label	Cluster size	T -statistics	MN	I Coo	rdinates
			x	y	Z
Color vs. Price and Year					
Frontal Inf Tri L	1531	6.37	-34	38	12
Precuneus R	1312	5.51	8	-56	44
Occipital Mid L	171	5.21	-36	-80	14



FIGURE 3.4: Brain regions with stronger activation in response to color vs. price and year.

3.3.2 ROI-Analysis Results: Brain Activity in the vmPFC

In the analyses of common and specific brain activity regions, significant activation in vmPFC was not observed, so ROI analysis was performed for the purpose of identifying differences among each choice in the vmPFC. The ROI analysis used the Brodmann area 10,14, 25, and 32 defined by Finger [68] as the region of vmPFC in the PickAtlas toolbox [35]. Figure 3.5 shows the significant activation in vmPFC during price, color, and year choice from the ROI analysis. The regions with strong activation in each condition are projected in the three views and the color intensity indicates the brain activity strength. These peak coordinates were given in Table 3.4 (p < 0.001 uncorrected). Significant activation was observed on vmPFC during each choice. In the price choice, a cluster with strong activation was observed (shown by the red circles in the left part of Figure 3.5). In the color choice, two clusters with strong activation were observed (shown by red circles and blue circles in the middle part of Figure 3.5). In the year choice, a cluster with strong activation was observed (shown by the green circles in the right part of Figure 3.5). The color of the circle in Figure 3.5 is classified according to the approximate coordinates of the activation region in vmPFC, and strong activation with similar coordinates indicated by the red circle was observed in the price and color conditions.

This result shows the possibility that detailed regions with activation on the vmPFC differ by type of evaluation criteria at the decision-making. In previous studies, there are many reports that vmPFC is involved in the decision-making or the preference selection tasks. O'Doherty et al. conducted evaluation tasks based on differences in the attractiveness of a human face. In this result, vmPFC had a significant activity when selecting a heterosexual face with a higher attractiveness [47]. McClure et al. reported that vmPFC activated significantly when participants drank a cola of more favourite brands in the case of drinking which had different preference degrees by them [44]. On the other hand, vmPFC had no activity in studies that were evaluation tasks due to differences in aesthetic preferences for paintings by Vartanian et al [43, 70]. The common point of these studies, these results was brain activity during decision-making for alternatives with different attractiveness or value based on a single evaluation criterion. The results of our study are the brain

activity by the different evaluation criteria, not the brain activity by the differences in the value of each evaluation criterion.



FIGURE 3.5: Brain regions with strong activation on the vmPFC.

TABLE 3.4: Significant brain activation on the vmPFC during price,
color, and year choice

Evaluation Criterion	T-statistics	MN	I Coc	ordinates
		x	у	Z
Price	7.12	6	20	42
Color	7.60	6	24	42
	6.10	30	54	6
Year	4.56	-10	50	0

3.3.3 Multi-voxel Pattern Analysis Results: Brain Regions Involved in Decision-Making

The precision results for the binary classification in vmPFC, insula, and NAcc are shown in Figure 3.6, 3.7, 3.8 and Table 3.5. In these figures, the color bars represent the average value of all 24 participants, and the error bars represent standard errors. The x-axis shows a combination of each binary classification (e.g., p-c shows the result of binary classification by price and color). In the precision using the vmPFC as an ROI, the highest average accuracy was 68.40% in the dummy-color binary classification, and the lowest was 58.51% in the price-color binary classification. Regarding the result of the insula as ROI, the highest average accuracy was 52.08% in the price-dummy

Chapter 3. Commonality and Specificity of Brain Activity Patterns Represented by the Evaluation Criteria

binary classification. Regarding the result of the NAcc as an ROI, the highest average accuracy was 63.37% in the dummy-color binary classification, and the lowest was 58.51% in the price-dummy binary classification.



FIGURE 3.6: Correct answer rate in the vmPFC



FIGURE 3.7: Correct answer rate in the insula



FIGURE 3.8: Correct answer rate in the NAcc

Regions of interest				Comb	inatio	ns of e	valuat	ion cri	teria			
	p-c	c-p	p-y	y-p	p-d	d-p	c-y	y-c	c-d	d-c	y-d	d-y
vmPFC												
Average accuracy [%]	58.5	61.5	62.2	64.4	61.5	61.5	67.7	61.1	64.2	68.4	66.0	62.7
\pm standard error	5.3	4.0	4.4	4.3	4.7	4.8	4.2	4.7	4.4	4.1	5.0	4.1
Insula												
Average accuracy [%]	56.0	58.3	52.8	54.7	52.1	54.3	57.1	55.9	57.5	57.3	55.9	57.6
\pm standard error	4.0	4.5	4.3	4.4	4.5	4.5	4.3	4.4	4.8	4.7	5.4	5.4
NAcc												
Average accuracy [%]	63.0	63.0	61.6	61.5	58.5	59.0	60.4	59.2	60.6	63.4	62.7	63.0
\pm standard error	4.4	4.0	4.6	4.4	5.3	5.4	4.8	5.0	4.5	4.6	5.2	4.9

Ξ.
ïf€
Ы
Ē
<u>.</u>
at
Ľ
a
6
Ч
ac
ē
ы
e,
Ξ
Je l
2
on
÷Ē
ca
Ξ
SS
la
ς,
Ŋ
ця.
Ξ
7
Ę,
ŝ
ē
Ę
Ч.
L L
g
Ĕ
ita
50
Ц Ц
5
es
. <u>D</u>
ra
g
ğ
0
š
ra
ve
Ā
ц)
3
Ľ
ΔB
Γ
- E - C

3.4 Discussion

3.4.1 Common and Specific Brain Regions Related to Decision-Making

This study investigated brain activation during the decision-making process of preference while using different evaluation criteria, and identified common and specific activation regions related to each evaluation criterion. As for common activation regions, the bilateral occipital gyri had significant activation during price choice, color choice, and year choice. The occipital gyrus is thought to be the brain region related to visual processing. Conversely, as for specific activation regions, the left fusiform gyrus was activated significantly during color choice. The fusiform gyrus is thought to be the brain region related to color processing [71]. Moreover, Oshin et al. reported that these regions are related to preference-rated tasks [70]. They concluded that the differential patterns of activation observed in bilateral occipital gyri and bilateral fusiform gyri in response to preference ratings are specific examples of their roles in evaluating reward-based stimuli that vary in emotional valence. However, this study investigated common brain activity by type of evaluation criteria. There is no report that the occipital gyrus and fusiform gyrus are involved in the influence on brain activity due to the difference in evaluation criteria, these results made the relationship between these gyrus functions and decision-making clearer. During color choice, the left insula was also activated. The insula is known for its involvement in value-based decision-making. It integrates the internal state, sensory signals, information about the salience, and relative value of stimuli during response selection [72]. Furthermore, the right precuneus was found to be more active during color choice compared with price and year choice. The precuneus is especially documented for its involvement in attention [73]. This suggests that the increased activation in the precuneus reflects increased attention to the color pairs compared with the price or year pairs. Therefore, it is possible that the participants made a simple numerical value comparison between the left information and the right information.

In the case of color choice, it is thought that the two color names displayed on the screen were processed based on the preference of the subjects. Therefore, it is considered that attention to color choice was higher than price and year choice, and the precuneus was activated.

The vmPFC has been shown in several neuroimaging studies to be active for a variety of primary and abstract rewards. These findings suggest that the vmPFC in the representation of complex appetitive states has several roles. However, all decision-making studies did not report that vmPFC is involved in decision-making. It suggests that the activated region differs depending on the evaluation criteria in the case of evaluating based on a value of alternative or preference by the individual. In this study, the analysis was performed based on our hypothesis that brain activity difference by evaluation criteria appears in vmPFC. Our results show that vmPFC was activated in all of the tasks by different evaluation criteria during the decision-making process. From the investigation of different activations on vmPFC for each evaluation criterion of alternatives, there were no differences in activation levels in the choice of each category. However, the activated detail regions on the vmPFC varied by each criterion. Some investigators have suggested that subjects' preference judgments stem result from "a competition between hedonic and utilitarian aspects of each choice alternative [56]", "competition between subjective emotional states such as desire and willpower [57]", "the selection of a specific dimension that enhances the contrast between the alternatives [58]", or "habit-based processing [59]". Therefore, the brain performed different processes with price, colour, and year choice. As a result, it is believed that choices by each alternative category represented brain activation on specific coordinates.

3.4.2 Brain activation patterns in the decision-making

The present study investigated the neural substrates associated with assessments of different criteria in decision-making. To this end, the study participants were presented with a pair of alternatives belonging to a single evaluation criterion and asked which product they wanted to choose based on evaluation criteria. We evaluated whether decision-making based on different evaluation criteria could be discriminated against based on the spatial activity pattern in different brain regions.

Considering the influence of the context of the vmPFC, insula, and NAcc activities observed in past studies, the ROIs of those areas were used. The vmPFC is active for a variety of primary and abstract rewards in several neuroimaging studies [56–59]. These findings suggest that the vmPFC plays several roles in the representation of complex choices, which suggests that the activated region differs depending on the evaluation criteria when based on an alternative value or preference by each individual. The insula has shown the possibility of triggering activation for the price during a purchasing decision, and the NAcc of inducing activation for individual preferences [60]. These findings suggest that the brain frames a preference as a potential benefit and price as a potential cost, thereby lending credence to the notion that consumer purchasing reflects an anticipatory combination of preference and price considerations. Although few studies have investigated the influence of differences in evaluation criteria on the brain, numerous studies have examined the influence of preferences and pleasure/discomfort when participants evaluate two or more stimuli as a decision-making task on the brain. These experimental designs are roughly divided into two types. The first is a design that presents a single stimulus in order, a design that the subject subsequently carries out in a single evaluation. The second presents a pair of stimuli at the same time, and the subjects select what they prefer more. In the present experiment, the latter design was adopted, in which the subjects evaluated a pair of alternatives belonging to a certain evaluation standard through comparisons. Numerous studies on decision-making have presented

a pair of images or character strings and performed evaluations, while others have used an experimental design that presents the same image and character strings of different contents, such as the present experimental design. Some studies have also reported results regarding activity in the vmPFC, insula, and NAcc [56, 57, 59].

As for the average accuracy of binary classification by MVPA, the present results using vmPFC, insula, and NAcc as ROIs exceeded the chance level. In the insula, even though the chance level exceeded the average precision, several combinations' standard error of the binary classification accuracy was lower than the chance level. The average classification accuracy in the insula was lower than those in the vmPFC and NAcc for all combinations of evaluation criteria. The average classification accuracy in the NAcc was about 60 %, and no difference in classification accuracy was seen for any evaluation criterion. On the other hand, vmPFC showed a difference in the binary classification accuracy for each combination of evaluation criteria. In the binary classification of price and year, the year was 2.26 % more accurate. The color was 2.95 % more accurate than the price and 6.60 % more accurate than the year. Although none of the differences in average accuracy was significant, classification in the vmPFC showed the possibility of being most classifiable when color was used as the evaluation criterion. These results suggest that brain regions involved in the decision and the preference, such as vmPFC and NAcc, represent differences in brain activity patterns in comparative decision-making.

Several studies applying MVPA to brain activity during decision-making have reported the following results. Bonnici et al. analyzed the brain pattern of the hippocampal in a simple decision-making task involving two highly similar scenes. They predicted which stimuli were being perceived under conditions of both perceptual certainty and ambiguity [74]. The classifier accuracy values of four regions on the hippocampus were above 50% of the chance level. Moreover, accuracies under the certainty condition were lower than the ambiguity condition, around 50-60%. This result has suggested that the comparison process may be occurring to reach a decision. The alternative pairs on our choice task were unambiguous. Although the brain regions of interest were different, the classification accuracies of our results were very similar to their results under the condition of certainty. Purchasing decisions are made under both certainty and ambiguous conditions. To expand the knowledge obtained in this study, it is necessary to pay attention to the ambiguity between alternative pairs in the future. Jai et al. conducted an evaluation and bidding task for apparel products under three visual presentation conditions (static picture, zooming, or model rotation videos) and analyzed the brain patterns in "a buy decision" and "a not buy decision" [75]. By the whole-brain classification analysis, the classifier accuracy rates were 95% in the zooming and rotation conditions, while the static condition had 75%. By the searchlight classification, there were some ROIs exclusively referred by certain presentations. These results suggest that the brain regions to which various information corresponds exhibit characteristic activity patterns, and that appropriate recognition of these patterns enables classification. On the other hand, Kim et al. investigated the relationship between purchase intention and perceived garment fit when purchasing decision-making [76]. They analyzed the brain activity during the task, including the phase of evaluating the fit of the model wearing clothes and the phase of purchasing consideration with the price displayed on the clothes. As a result of the MVPA for whole brain searchlight, the classification accuracies were more than 50% of the chance level at 11 brain regions. Among them, the superior parietal lobule exceeded 80% with the highest accuracy. One of the reasons for these accuracies higher than our results may be that their tasks included multiple evaluation criteria such as garment fit and price. Our results showed that decision-making based on individual criteria is represented in brain patterns. Still, we believe that decision-making based on multiple criteria may represent more characteristic brain patterns.

This study focused on customer preferences based on individual evaluation criteria. Although the results of the classifier did not show activity patterns in specific brain regions that depended on the evaluation criteria, they indicated the possibility that preference-based decision-making could be represented in brain patterns. It is necessary to process various factors such as reward, risk, and strategy in decision-making. In particular, regarding purchasing decisions, it is said that the factor of individual preferences produces better decisions [77–80]. The development of this research is expected to help elucidate the neural basis of decision-making and to contribute to selection, online shopping, or marketing strategies.

Chapter 4

Conclusion

In this dissertation, I described the three investigated results which are learning and representation of planning ability in the human brain involving executive function and decision-making.

Firstly, we focused on brain activity before and after learning. The effect of the planning and decision-making process concerning executive function on brain activity via long-time learning was investigated. The experimental design involved long-time learning in examining differences in cerebral activity. The participants performed individualized learning experiments involving the ToH to identify brain regions. As a result, brain regions involved in executive functions showed differences in activity between before and after learning. This study focused on the DLPFC, parietal lobe, inferior frontal gyrus, and premotor cortex as brain regions involved in executive function. Activity in these brain regions declined compared with before learning. The results indicated a conflicting trend between learning progress and brain activity.

The ability to solve the ToH requires task anticipation, planning, and decisionmaking. In this experiment, participant learning progressed, and their learning curves converged. Therefore, their problem-solving and task-anticipation abilities improved. Moreover, the time required for the planning and decision-making process was simplified. It was also clear that the degree of activation tends to decrease with the number of tasks, such as through M2 to M3. The frontal pole, or the PFC, is related to the ToH task.

On the other hand, in the present long-time experiment, activation differences were obtained at the DLPFC, parietal lobe, inferior frontal gyrus, and premotor cortex after learning in each participant. We believe these activation differences were due to progressive learning about executive function.

This experiment was conducted for a long-time with only four participants due to logistic issues. Therefore, the conclusions driven by this research are not generic. More participants should be considered for sound and generalized conclusions. However, having more participants in this type of research is very challenging.

Nevertheless, we considered that the area of brain activation changes each time during long-time learning. This study analyzed behaviors and brain activities in M1–M3 in long-time learning. Exploring the dynamics of brain activity in long-time learning would also be essential. The learning curve for behavioral performance indicated this possibility.

Furthermore, a relationship with the dynamics of brain activities was observed. Various abilities are needed to solve the ToH task. The participants' learning progress from learning curves was observed during the experiments. It can be stated that the observed learning progress implies improving overall solution ability. Therefore, the dynamics of brain activity, which are associated with this ability to contribute more, should be investigated in a future study.

Next, we focused on brain activation during the decision-making process, which was selected based on a single evaluation criterion and by conducting fMRI experiments. The experiment measured brain activity during a paired comparison based on the price, color, or product year as a single evaluation criterion by fMRI, in order to elucidate the brain regions comprehensively involved in decision-making and to identify the specific brain regions involved in each evaluation criterion. From the scanned fMRI data of the choice tasks in each evaluation criterion, common brain activation and specific brain activation were performed by whole-brain analysis. The bilateral occipital gyri had significant activation as a common activation region in choice tasks by each evaluation criterion. This result had shown that the difference in preference rate reflected the brain activity difference in the occipital gyrus.

This study investigated the common brain activity by type of the evaluation criteria, occipital gyrus was significantly activated at decision-making by all evaluation criteria. Therefore, this result suggests that the occipital gyrus is a region comprehensively related to decision-making. As for specific activation regions in choice tasks by each evaluation criterion, the left fusiform gyrus, left insula, and right precuneus was activated significantly during color choice. During price and year choice, specific activation was not observed. The results suggest that attention to color choice becomes greater compared to price or year choice. With the vmPFC, which has been reported to be involved in decision-making, significant activation was not observed from the results of the whole-brain analysis. The tasks of previous reports are decision-making for alternatives with different attractiveness or values based on a single evaluation criterion.

This study analyzed the brain activation differences among each evaluation criterion on decision-making, so these results did not focus on the differences in attractiveness or values. An ROI analysis for vmPFC was performed in order to confirm the differences the brain activation in the vmPFC by evaluation criteria. In the ROI analysis result, significant activations on the vmPFC were observed with all tasks by each evaluation criterion, and these detailed coordinates were located in different coordinates. This result suggests a possibility that detailed activity regions in vmPFC differ by the types of evaluation criteria.

Finally, we investigated the representation of evaluation criterion categories in decision-making using fMRI and MVPA. Price, color, and year were used as the evaluation criteria. We focused on the vmPFC, NAcc, and insula as ROIs.

Each combination of the four evaluation criteria was analyzed into a binary classification by MVPA. From the binary classification accuracy obtained from MVPA, we evaluated the regions that reflected differences in evaluation criteria among the ROIs. From the results of the binary classification by MVPA, the vmPFC, and NAcc showed that these regions were capable of expressing the influence of the evaluation criteria during decision-making.

In this research, we applied classification analysis from brain activity by adopting price, color, and year as evaluation criteria because they are often involved in purchasing decisions for various products. In actual purchase decision-making, consumers also refer to various evaluation criteria. In a future study, it will be necessary to investigate brain activity patterns expressed by evaluation criteria other than those adopted in the present study and to verify whether they can be classified similarly.

References

- Muriel D Lezak. "The problem of assessing executive functions". In: *International journal of Psychology* 17.1-4 (1982), pp. 281–297.
- [2] Julie A Alvarez and Eugene Emory. "Executive function and the frontal lobes: a meta-analytic review". In: *Neuropsychology review* 16.1 (2006), pp. 17–42.
- [3] Sanjeev Swami. "Executive functions and decision making: A managerial review". In: *IIMB Management Review* 25.4 (2013), pp. 203–212.
- [4] Dan Ariely and Gregory S Berns. "Neuromarketing: the hope and hype of neuroimaging in business". In: *Nature reviews neuroscience* 11.4 (2010), pp. 284–292.
- [5] Hilke Plassmann, J. O'Doherty, and Antonio Rangel. "Orbitofrontal Cortex Encodes Willingness to Pay in Everyday Economic Transactions". In: *Journal* of Neuroscience 27.37 (2007), pp. 9984–9988.
- [6] Isabella M. Kopton and Peter Kenning. "Near-infrared spectroscopy (NIRS) as a new tool for neuroeconomic research". In: *Frontiers in Human Neuroscience* 8.August (2014), pp. 1–13.
- [7] M. Andrea Pisauro et al. "Neural correlates of evidence accumulation during value-based decisions revealed via simultaneous EEG-fMRI". In: *Nature Communications* 8.May (2017), p. 15808.
- [8] Yukiyasu Kamitani and Frank Tong. "Decoding the visual and subjective contents of the human brain." In: *Nature neuroscience* 8.5 (2005), pp. 679–685.
- [9] Kenneth A. Norman et al. "Beyond mind-reading: multi-voxel pattern analysis of fMRI data". In: *Trends in Cognitive Sciences* 10.9 (2006), pp. 424–430.

- [10] Junichi Chikazoe et al. "Population coding of affect across stimuli, modalities and individuals". In: *Nature Neuroscience* 17.8 (2014), pp. 1114–1122.
- [11] Jordan Grafman et al. "Cognitive planning deficit in patients with cerebellar atrophy". In: *Neurology* 42.8 (1992), pp. 1493–1493.
- [12] Miguel Angel Guevara et al. "Prefrontal-parietal correlation during performance of the towers of Hanoi task in male children, adolescents and young adults". In: *Developmental Cognitive Neuroscience* 2.1 (2012), pp. 129–138.
- [13] Geoff Ward & Alan Allport. "Planning and problem solving using the five disc Tower of London task". In: *The Quarterly Journal of Experimental Psychology: Section A* 50.1 (1997), pp. 49–78.
- [14] John Karat. "A model of problem solving with incomplete constraint knowledge". In: *Cognitive Psychology* 14.4 (1982), pp. 538–559.
- [15] John R Anderson, Mark V Albert, and Jon M Fincham. "Tracing problem solving in real time: fMRI analysis of the subject-paced Tower of Hanoi". In: *Journal* of cognitive neuroscience 17.8 (2005), pp. 1261–1274.
- [16] Rebecca Bull, Kimberly Andrews Espy, and Theresa E Senn. "A comparison of performance on the Towers of London and Hanoi in young children". In: *Journal of Child Psychology and Psychiatry* 45.4 (2004), pp. 743–754.
- [17] Jon M Fincham et al. "Neural mechanisms of planning: a computational analysis using event-related fMRI". In: *Proceedings of the National Academy of Sciences* 99.5 (2002), pp. 3346–3351.
- [18] Kenneth Kotovsky, John R Hayes, and Herbert A Simon. "Why are some problems hard? Evidence from Tower of Hanoi". In: *Cognitive psychology* 17.2 (1985), pp. 248–294.
- [19] Sharlene D Newman et al. "Frontal and parietal participation in problem solving in the Tower of London: fMRI and computational modeling of planning and high-level perception". In: *Neuropsychologia* 41.12 (2003), pp. 1668–1682.

- [20] Terrence Stewart and Chris Eliasmith. "Neural cognitive modelling: A biologically constrained spiking neuron model of the Tower of Hanoi task". In: Proceedings of the Annual Meeting of the Cognitive Science Society. Vol. 33. 33. 2011.
- [21] Marilyn C Welsh and Mariëtte Huizinga. "Tower of Hanoi disk-transfer task: Influences of strategy knowledge and learning on performance". In: *Learning* and Individual Differences 15.4 (2005), pp. 283–298.
- [22] D.T. Stuss et al. "Language functioning after bilateral prefrontal leukotomy". In: *Brain and Language* 28.1 (1986), pp. 66–70.
- [23] Alain Dagher et al. "Mapping the network for planning: a correlational PET activation study with the Tower of London task". In: *Brain* 122.10 (1999), pp. 1973– 1987.
- [24] Richard HC Lazeron et al. "Visualizing brain activation during planning: the tower of London test adapted for functional MR imaging". In: *American Journal* of Neuroradiology 21.8 (2000), pp. 1407–1414.
- [25] Adrian M Owen et al. "Planning and spatial working memory: a positron emission tomography study in humans". In: *European Journal of Neuroscience* 8.2 (1996), pp. 353–364.
- [26] Ulrich Schall et al. "Functional brain maps of Tower of London performance: a positron emission tomography and functional magnetic resonance imaging study". In: *Neuroimage* 20.2 (2003), pp. 1154–1161.
- [27] Odile A Van Den Heuvel et al. "Frontostriatal system in planning complexity: a parametric functional magnetic resonance version of Tower of London task". In: *Neuroimage* 18.2 (2003), pp. 367–374.
- [28] Stuartk Card, THOMASP MORAN, and Allen Newell. "The model human processor- An engineering model of human performance". In: *Handbook of perception and human performance*. 2.45–1 (1986).

- [29] Odile A Van Den Heuvel et al. "Frontal-striatal dysfunction during planning in obsessive-compulsive disorder". In: Archives of general psychiatry 62.3 (2005), pp. 301–309.
- [30] Eli Vakil and Eyal Heled. "The effect of constant versus varied training on transfer in a cognitive skill learning task: The case of the Tower of Hanoi Puzzle". In: *Learning and Individual Differences* 47 (2016), pp. 207–214.
- [31] Rachel Schiff and Eli Vakil. "Age differences in cognitive skill learning, retention and transfer: The case of the Tower of Hanoi Puzzle". In: *learning and individual differences* 39 (2015), pp. 164–171.
- [32] Sharlene D Newman, John A Greco, and Donghoon Lee. "An fMRI study of the Tower of London: a look at problem structure differences". In: *Brain research* 1286 (2009), pp. 123–132.
- [33] Sally Ozonoff and Jenise Jensen. "Brief report: Specific executive function profiles in three neurodevelopmental disorders". In: *Journal of autism and developmental disorders* 29.2 (1999), pp. 171–177.
- [34] Wei Xing Toh, Hwajin Yang, and Andree Hartanto. "Executive function and subjective well-being in middle and late adulthood". In: *The Journals of Gerontology: Series B* 75.6 (2020), e69–e77.
- [35] Joseph A Maldjian et al. "An automated method for neuroanatomic and cytoarchitectonic atlas-based interrogation of fMRI data sets". In: *NeuroImage* 19.3 (2003), pp. 1233–1239.
- [36] Joseph A Maldjian, Paul J Laurienti, and Jonathan H Burdette. "Precentral gyrus discrepancy in electronic versions of the Talairach atlas". In: *NeuroIm*age 21.1 (Jan. 2004), pp. 450–455.
- [37] MW Schlund and D Ortu. "Experience-dependent changes in human brain activation during contingency learning". In: *Neuroscience* 165.1 (2010), pp. 151– 158.

- [38] Katsuyuki Sakai et al. "Transition of brain activation from frontal to parietal areas in visuomotor sequence learning". In: *Journal of Neuroscience* 18.5 (1998), pp. 1827–1840.
- [39] MM Sohlberg. "Management of dysexecutive symptoms". In: *Cognitive Rehabilitation, An Integrative Neuropsychological Approach* (2001), pp. 230–268.
- [40] Raphael M Bonelli and Jeffrey L Cummings. "Frontal-subcortical circuitry and behavior". In: *Dialogues in clinical neuroscience* 9.2 (2007), p. 141.
- [41] Kenoki Ohuchida et al. "The frontal cortex is activated during learning of endoscopic procedures". In: *Surgical endoscopy* 23.10 (2009), pp. 2296–2301.
- [42] Hyunkyu Lee et al. "Videogame training strategy-induced change in brain function during a complex visuomotor task". In: *Behavioural Brain Research* 232.2 (2012), pp. 348–357.
- [43] Susanne Erk et al. "Cultural objects modulate reward circuitry." In: *Neuroreport* 13.18 (Dec. 2002), pp. 2499–2503.
- [44] Samuel M. McClure et al. "Neural correlates of behavioral preference for culturally familiar drinks". In: *Neuron* 44.2 (2004), pp. 379–387.
- [45] Edmund T. Rolls, Morten L. Kringelbach, and Ivan E.T. De Araujo. "Different representations of pleasant and unpleasant odours in the human brain". In: *European Journal of Neuroscience* 18.3 (2003), pp. 695–703.
- [46] Hilke Plassmann et al. "Marketing actions can modulate neural representations of experienced pleasantness". In: *Proceedings of the National Academy of Sciences* 105.3 (2008), pp. 1050–1054.
- [47] J. O'Doherty et al. "Beauty in a smile: The role of medial orbitofrontal cortex in facial attractiveness". In: *Neuropsychologia* 41.2 (Jan. 2003), pp. 147–155.
- [48] Brian Knutson et al. "A region of mesial prefrontal cortex tracks monetarily rewarding outcomes: Characterization with rapid event-related fMRI". In: *NeuroImage* 18.2 (2003), pp. 263–272.

- [49] Brian Knutson et al. "Neural predictors of purchases". In: *Neuron* 53.1 (2007), pp. 147–156.
- [50] Kaspar Meyer and Jonas T Kaplan. "Cross-Modal Multivariate Pattern Analysis". In: *Journal of Visualized Experiments* 57.57 (2011), p. 3307.
- [51] Konstantinos Katsikopoulos and Marc Canellas. "Decoding human behavior with big data? Critical, constructive input from the decision sciences". In: AI Magazine 43.1 (2022), pp. 126–138.
- [52] Y Yau et al. "Neural Correlates of Evidence and Urgency During Human Perceptual Decision-Making in Dynamically Changing Conditions". In: *Cerebral Cortex* 30.10 (June 2020), pp. 5471–5483.
- [53] Zhiyuan Liu et al. "Dissociating value-based neurocomputation from subsequent selection-related activations in human decision-making". In: *Cerebral Cortex* 32.19 (Jan. 2022), pp. 4141–4155.
- [54] Chiranji Lal Chowdhary et al. "An Efficient Segmentation and Classification System in Medical Images Using Intuitionist Possibilistic Fuzzy C-Mean Clustering and Fuzzy SVM Algorithm". In: Sensors 20.14 (2020).
- [55] Muhammad Rizwan et al. "Brain Tumor and Glioma Grade Classification Using Gaussian Convolutional Neural Network". In: *IEEE Access* 10 (2022), pp. 29731– 29740.
- [56] Ravi Dhar and Klaus Wertenbroch. "Consumer choice between hedonic and utilitarian goods". In: *Journal of Marketing Research* 37.February (2000), pp. 60– 71.
- [57] Stephen J. Hoch and George F. Loewenstein. "Time-inconsistent Preferences and Consumer Self-Control". In: *Journal of Consumer Research* 17.4 (Mar. 1991), pp. 492–507.
- [58] BA Mellers and Shi-Jie Chang. "Preferences, prices, and ratings in risky decision making." In: *Journal of experimental psychology* 18.2 (1992), pp. 347–361.
- [59] Eva Lindbladh and Carl Hampus Lyttkens. "Habit versus choice: The process of decision-making in health-related behaviour". In: *Social Science and Medicine* 55.3 (Aug. 2002), pp. 451–465.
- [60] Nikolaas N. Oosterhof, Andrew C. Connolly, and James V. Haxby. "CoSMoMVPA: Multi-Modal Multivariate Pattern Analysis of Neuroimaging Data in Matlab/GNU Octave". In: Frontiers in Neuroinformatics 10 (July 2016), p. 27.
- [61] Mehrbakhsh Nilashi et al. "Neuromarketing: a review of research and implications for marketing". In: *Journal of Soft Computing and Decision Support Systems* 7.2 (2020), pp. 23–31.
- [62] Aida Azlina Mansor and Salmi Mohd Isa. "Fundamentals of neuromarketing: What is it all about?" In: *Neuroscience Research Notes* 3.4 (2020), pp. 22–28.
- [63] AHMED H Alsharif et al. "Consumer behaviour through neuromarketing approach". In: *Journal of Contemporary Issues in Business and Government* 27.3 (2021), pp. 344–354.
- [64] J. Bradley C Cherry et al. "Neurofunctional correlates of ethical, food-related decision-making". In: *PLoS ONE* 10.4 (2015).
- [65] Lawrence L Garber, Raymond R Burke, and J Morgan Jones. "The role of package color in consumer purchase consideration and choice". In: (2000).
- [66] Karl J Friston et al. "Statistical parametric maps in functional imaging: A general linear approach". In: *Human Brain Mapping* 2.4 (1994), pp. 189–210.
- [67] Chih-Chung Chang and Chih-Jen Lin. "LIBSVM: a library for support vector machines". In: ACM transactions on intelligent systems and technology (TIST) 2.3 (2011), pp. 1–27.
- [68] Elizabeth C Finger et al. "Abnormal ventromedial prefrontal cortex function in children with psychopathic traits during reversal learning". In: Arch Gen Psychiatry 65.5 (2008), pp. 586–594.

- [69] N. Tzourio-Mazoyer et al. "Automated anatomical labeling of activations in SPM using a macroscopic anatomical parcellation of the MNI MRI singlesubject brain". In: *NeuroImage* 15.1 (2002), pp. 273–289.
- [70] Oshin Vartanian and Vinod Goel. "Neuroanatomical correlates of aesthetic preference for paintings". In: *Neuroreport* 15.3 (2004), pp. 893–.
- [71] Jennifer M. Groh. "The tell-tale brain". In: *Journal of Clinical Investigation* 121.8 (Aug. 2011), pp. 2953–2953.
- [72] Martin P. Paulus and Murray B. Stein. *An Insular View of Anxiety*. Aug. 2006.
- [73] Andrea E. Cavanna and Michael R. Trimble. *The precuneus: A review of its functional anatomy and behavioural correlates.* Jan. 2006.
- [74] Heidi M Bonnici et al. "Multi-voxel pattern analysis in human hippocampal subfields". In: Frontiers in human neuroscience 6 (2012), p. 290.
- [75] Tun-Min Jai et al. "Seeing it is like touching it: unraveling the effective product presentations on online apparel purchase decisions and brain activity (An fMRI Study)". In: *Journal of Interactive Marketing* 53.1 (2021), pp. 66–79.
- [76] Hesun Erin Kim, Joon Hee Kwon, and Jae-Jin Kim. "Neural correlates of garment fit and purchase intention in the consumer decision-making process and the influence of product presentation". In: *Frontiers in Neuroscience* 15 (2021), p. 609004.
- [77] Itamar Simonson and Stephen M Nowlis. "The role of explanations and need for uniqueness in consumer decision making: Unconventional choices based on reasons". In: *Journal of Consumer Research* 27.1 (2000), pp. 49–68.
- [78] Ravi Dhar. "Consumer preference for a no-choice option". In: *Journal of con-sumer research* 24.2 (1997), pp. 215–231.
- [79] Dan Ariely. "Controlling the information flow: Effects on consumers' decision making and preferences". In: *Journal of consumer research* 27.2 (2000), pp. 233– 248.

[80] Paul Slovic. "The construction of preference." In: *American psychologist* 50.5 (1995), p. 364.

Publications

Journal

- Keita Mitani and Yukinobu Hoshino. "Investigation of Involvement between Specific Brain Regions and Evaluation Criteria Elements in a Two-Selections Task". In: *Computational Intelligence and Neuroscience*. 2022.3999223, (2022).
- Keita Mitani, Namal Rathnayake, Upaka Rathnayake, Tuan Linh Dang, and Yukinobu Hoshino. "Brain Activity Associated with the Planning Process during the Long-Time Learning of the Tower of Hanoi (ToH) Task: A Pilot Study". In: *Sensors.* 22.21, (2022), pp. 8283–8296.
- 3. 三谷 慶太, 星野 孝総. "長期学習者の成長曲線に基づく学習度合い予測モデルの開発と評価". 知能と情報. 33.4, (2021), pp. 845–859.
- Keita Mitani and Yukinobu Hoshino. "Verification for Commonality or Specificity of Brain Representations Related to the Different Evaluation Criteria". In: *International Journal of Innovative Computing Information and Control (IJI-CIC)*. 14.5 (2018), pp. 1553–1563.
- Yukinobu Hoshino and Keita Mitani. "A Proposal of a Usability Scale System for Rehabilitation Games Based on the Cognitive Therapeutic Exercise". In: *International Journal of Innovative Computing Information and Control (IJICIC)*. 14.4, (2018), pp. 1189–1205.

 6. 星野 孝総, 岡坂 翔, 三谷 慶太. "BOLD信号による賦活状態識別のためのロジス ティック関数を用いた差分進化法による識別器の最適化". 知能と情報. 28.3, (2016), pp. 617–626.

International Conference

- Ryosuke Hayasaka, Keita Mitani, and Yukinobu Hoshino. "Basic Verification of the brain Areas Related with the Time Measurement to use BCI". In: *Proceedings of Joint 17th World Congress of International Fuzzy Systems Association* (IFSA) and 9th International Conference on Soft Computing and Intelligent Systems (SCIS). (2017), pp. 27-30.
- 2. Ryosuke Hayasaka, Keita Mitani, and Yukinobu Hoshino. "Basic Verification of the Brain Areas Involved in the Time Perception to use BCI". In: *International Symposium on Interaction Design and Human Factors (IDHF)*. (2016).
- Yuya Sano, Keita Mitani, and Yukinobu Hoshino. "Development of fMRI Synchronization Signal Generator Using Arduino for Pseudo Experimental". In: *International Symposium on Interaction Design and Human Factors (IDHF)*. (2016).
- Keita Mitani and Yukinobu Hoshino. "Correlation of Brain Activity and Questionnaire at the Decision-Making". In: *Joint 8th International Conference on Soft Computing and Intelligent Systems (SCIS) and 17th International Symposium on Advanced Intelligent Systems (ISIS)*. (2016), pp. 445-448.
- 5. Ryosuke Hayasaka, Keita Mitani, and Yukinobu Hoshino. "Workings of the Brain in the Time Perception Using the Time Guessing". In: *Joint 8th International Conference on Soft Computing and Intelligent Systems (SCIS) and 17th International Symposium on Advanced Intelligent Systems (ISIS).* (2016), pp. 929-931.

- 6. Keita Mitani and Yukinobu Hoshino. "Brain Activation Difference of Social Learning and Individual Learning by Planning Game". In: Proceedings of 16th International Symposium on Advanced Intelligent Systems(ISIS). (2015).
- Yukinobu Hoshino, Keita Mitani, Naoki Miura, Hiroki C. Tanabe, and Kenji Nagai. "fMRI Experiment About Executive Function by Difference of Learning Methods: Using Tower of Hanoi". In: *RNMH2014 The Second International Conference.* (2014).

Domestic Conference

- 三谷慶太, 星野孝総, "意思決定時の評価基準に関連する共通・特異的な脳領域の検証". 2018 IEEE SMC Hiroshima Chapter 若手研究会. (2018), pp. 132-135.
- 佐野 友哉, 三谷 慶太, 星野 孝総, "Arduinoを用いた疑似fMRI実験用装置の開発と検証". 第29回バイオメディカル・ファジィ・システム学会年次大会.
 (2016).
- 3. 早坂 涼佑, 三谷 慶太, 星野 孝総, "ヒトの自発的な時間計測に関与する脳部位の解明への試み". 第32回ファジィシステムシンポジウム. (2016).
- 三谷 慶太, 星野 孝総, "計画ゲームを用いた遂行機能タスクに関する脳賦活の 検証". 第31回ファジィシステムシンポジウム. (2015).
- 5. 三谷 慶太, 星野 孝総, 三浦 直樹, 田邊 宏樹, "学習方法の違いに関するfMRI解 析". 第30回ファジィシステムシンポジウム. (2014).
- 三谷 慶太, 越村 美里, 星野 孝総, "携帯電話の買い替え意思決定支援システムのためのFuzzyAHP法を用いた一考察". 第30回ファジィシステムシンポジウム. (2014).
- 7. 星野 孝総, 三谷 慶太, 三浦 直樹, 田邊 宏樹, 長井 謙治, "ハノイの塔を用いた社 会学習と個体学習差の検証実験". 第9回研究大会 ネアンデルタールとサピエ ンス交替劇の真相:学習能力の進化に基づく実証的研究. (2014).

Book

 Yukinobu Hoshino, Keita Mitani, Naoki Miura, Hiroki C. Tanabe, and Kenji Nagai. "Motion Analysis for Stone-Knapping of the Skilled Levallois Technique, Dynamics of Learning in Neanderthals and Modern Humans Volume 2: Cognitive and Physical Perspectives". *Springer.* (2014), pp. 79-90.